**Open Access** 

# SEED-AN and a non-specialised, severe mental illness (SMI) community treatment model: perspectives of professionals and patients of a QoL-focussed treatment

Laura Schut<sup>1,4\*</sup>, Marc Godschalk<sup>2</sup> and Martie de Jong<sup>3</sup>

# Abstract

**Background** Improving care for SEED-AN patients is urgently needed. Current specialist eating disorder (SEDU) programs have minimal impact, resulting in poor quality of life. Flexible assertive community treatment (FACT) focuses on improving the quality of life of people with severe mental illness.

**Aim** This study systematically examined professionals' and patients' experiences with existing care for SEED AN in two settings in two health districts in the Netherlands. Specifically, the specialised Eating Disorder Unit and FACT.

**Methods** A qualitative approach was used: Twelve professionals participated in focus groups, and four professionals and six patients were interviewed individually. Reflective thematic analysis was used to analyse the data.

**Results** Four themes were constructed: "We *feel ignorant in the treatment of SEED-AN patients"; "There is a disability to act"; "We are more than executive practitioners alone"; and "The professionals at FACT have given me back my confidence in treatment".* The findings reflect how support for SEED-AN patients is provided by both FACT and SEDU professionals and how SEED-AN patients experience the FACT approach.

**Discussion** A resource group model promotes collaboration among professionals, patients and the patient's support system. The model is expected to increase knowledge about SEED-AN among stakeholders who support the quality of life of SEED-AN patients.

# **Plain English summary**

Of the group of patients diagnosed with anorexia nervosa, approximately 20–25% develop a severe and persistent eating disorder - anorexia nervosa (SEED-AN). Families, spouses and carers struggle with a range of emotions and challenges associated with this form of the disease. Current specialised programmes often focus on the recovery of younger people. In the Netherlands, people with severe mental illness can be supported by specialised outpatient

\*Correspondence: Laura Schut I.moerland-schut@youz.nl

Full list of author information is available at the end of the article



© The Author(s) 2024. **Open Access** This article is licensed under a Creative Commons Attribution-NonCommercial-NoDerivatives 4.0 International License, which permits any non-commercial use, sharing, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons licence, and indicate if you modified the licensed material. You do not have permission under this licence to share adapted material derived from this article or parts of it. The images or other third party material in this article are included in the article's Creative Commons licence, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons licence and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this licence, visit http://creativecommons.org/licenses/by-nc-nd/4.0/.

teams. These teams are often trained in the Flexible Assertive Community Treatment (FACT) method. In addition to treating the disorder, the method focuses on improving quality of life in several areas, such as social participation, work, independent living and leisure. Patients with SEED-AN are often excluded from such programmes, but could benefit from the method. With this study, we have gained a better understanding of why most of the patients with SEED-AN have not been able to use these services. Both professionals and SEED-AN patients shared their experiences and opinions. Our findings reveal a number of challenges, such as the unfamiliarity of professionals with eating disorders, but also the perceptions of SEED-AN patients who do receive this support. Collaboration between patients, professionals, family members and other supporters was considered crucial. our recommendation is network development, the use of a resource group model appeared to be helpful in this.

Keywords Anorexia nervosa, SEED-AN, Reflexive thematic analysis, FACT, Resource group model

# Background

Over a period of three decades, the number of cases of anorexia nervosa in the Netherlands has remained stable; nevertheless, during the COVID-19 pandemic, the diagnostic incidence of anorexia in Western countries was higher in 2020 and steadily increased from March 2020 to 1.5 by the end of that year [1]. However, there seems to be an improvement in the recognition of eating disorders, possibly through greater public awareness and better diagnostic tools [2]. A significant proportion of people diagnosed with anorexia nervosa develop a longer duration of the disorder, despite treatment according to guidelines [3]. These patients reported negative effects of therapy and admission to a psychiatric institution [4]. Unfortunately, the risk of dying because of an eating disorder is high [5, 6]. Recent research has shown that the mortality rate of people with AN can be up to five times higher [2], with one in five people dying by suicide [5]. Approximately half of people with an eating disorder experience partial recovery [7-9]. A Dutch study [3] described partial recovery in adolescents on the basis of a reduction in clinical symptoms, such as BMI, degree of functional disability and need for supervision in anorexia nervosa specifically, as described in the Eating Disorder Examination (EDE) [10, p. 265–308]. However, 30% continue to have one or more symptoms, and 20-25% have a long-lasting form of anorexia nervosa in which both physical and psychological complications have a negative impact on quality of life [7, 11, 12].

There is also an ongoing debate about the definition of severe and enduring eating disorders [4]. A recent metaanalysis posed alternative conceptualisations of the issue, considering treatment options and complex intra- and interpsychic processes of SEED-AN [13]. In our study, we use the term SEED-AN to exclude other types of longlasting eating disorders to improve the approach [14]. Owing to a lack of international consensus, this study used the criteria in the Dutch Eating Disorders Standard; individuals are considered to have a chronic illness if AN is present continuously for 5 years or intermittently for 10 years [15].

Living with SEED-AN can have a serious impact on an individual's quality of life, leading to psychological problems such as depression and anxiety, as well as problems with memory, concentration, eating, weight and appearance [16]. The poor nutritional status associated with anorexia can lead to serious physical complications, such as heart problems, irritation of the bowel and esophagus, and erosion of tooth enamel [14, 17]. In addition, living with SEED-AN has a negative effect on an individual's social life [18]. Patients may feel that they are a burden to those around them and to healthcare providers [18-22]. Many patients are unable to study or work, which can lead to social isolation. As a result, they become dissatisfied with their lives in many areas, including health, sexuality, finances, leisure and social status. Some patients therefore live in poor physical, psychological and social conditions [18, 22, 23]. A quotation published in 2019 illustrates this:

# "Dear friends,

No one has failed to notice that my health is getting worse, despite my attempts to feel better in all sorts of ways. Sometimes I seem to have found the solution, but 10 steps forward always seem to end up in 20 steps back." Sandy, 2019 [18, p. 35].

Overall, eating disorders are serious but treatable illnesses. Early treatment, such as family-based therapy (FBT) [24] for young people, has good outcomes. Although long-term outcomes are limited, early treatment reduces risk factors such as body dissatisfaction and low self-esteem but also promotes help-seeking behaviour and symptom recognition [25, 26].

Unfortunately, eating disorders are also the most common chronic disorders among young people [3]. The Health Council of the Netherlands has therefore recommended a unified national approach focusing on prevention and early treatment [27]. Evidence-based treatment methods are still limited; the Dutch guidelines [15] suggest an individualised plan that includes collaboration with family caregivers and/or expertise care. Current treatment options include psychological interventions, the treatment of physical complaints, pharmacotherapy, or a combination of these approaches. These treatments are available in a specialist setting (specialised eating disorder unit [SEDU]). Treatment for AN initially focused on restoring physical health and normalising weight and eating behaviour, and current treatment programmes address mainly the physical and psychological symptoms of patients with a more acute form of anorexia nervosa. These treatments seem to be more effective in younger patients with a shorter duration of illness; therefore, they do not fully address the needs of patients with SEED-AN [28, 29], namely, reducing persistent negative effects on quality of life [13, 14, 18, 23, 24].

In addition, SEDU professionals are increasingly confronted with the demand for nonspecific medical and sometimes palliative support from persons suffering from SEED-AN [30]. On the basis of the findings concerning who strongly influences quality of life, it seems reasonable that persons with SEED-AN require a different approach. Touyz et al. [22] adapted the Specialist Supportive Clinical Management Model (SSCM) [31, 32] to SEED-AN and compared the approach with CBT-E [10]. An adapted treatment paradigm was proposed; the focus should shift from complete recovery to improving quality of life and avoiding experiences of treatment failure [22, 33].

Approximately 20% of the persons diagnosed with AN remain chronically ill [9]). Unfortunately, there is no evidence-based treatment for SEED-AN [34]. Some SEDUs offer specialised help, such as group sessions, often led by a therapist and/or experienced individuals, with the aim of recognising and acknowledging the common issues (e.g., problems responsible for a diminished quality of life, physical healthcare issues) within SEED-AN [35]. Furthermore, clinical pathways are offered on the basis of SSCM [31, 32]. SSCM can improve motivation and quality of life to support recovery from the core pathology of anorexia. However, there is still insufficient knowledge about which SEED patients benefit from weight restoration versus focusing on quality of life alone. SSCM offers the opportunity to treat existing comorbidities with the primary aim of improving quality of life [36]. There are approximately twenty SEDUs in the Netherlands. These SEDUs offer a stepped-up, stepped-down strategy of care to all eating disorder patients, including those in the SEED-AN [1]. However, not every SEDU offers treatment to individuals with SEED-ANs. In addition, followup contact in the patient's own region after the end of treatment is often suggested. For patients who live a long way from an existing eating disorder programme, this follow-up process is not always possible if a suitable solution cannot be found. In addition, the physical condition of the SEED-AN patient also plays a role, which makes travelling to an SEDU challenging. Early findings from a recent study focusing on peer support were hopeful, and a reduction in feelings of isolation and hopelessness was observed [37]. Therefore, it seems logical to provide follow-up treatment after the end of SEDU contact through a regional outreach programme to prevent relapse and to work towards improving quality of life.

In the Netherlands, there is a separate mental health care system for people with severe mental illness (SMI). SMI is characterised by a persistent and severe mental disorder accompanied by limitations in social and societal functioning [38, 39]. A flexible assertive community treatment (FACT) approach has been developed and broadly implemented in the Netherlands [40]. In the Netherlands, one FACT team is responsible for the mental health care of an average of 200 SMI patients in a postcode area of approximately 50,000 people, focusing on goals related to physical, psychological, social and societal aspects as well as (partial) recovery and rehabilitation. FACT teams provide support in the person's own environment and social network [40]. The FACT method is a well-researched, evidence-based intervention that has been implemented in many countries [41]. FACT offers a similar philosophy, such as harm reduction approaches [42, 43], for instance, the HARMONI programme (HARm MinimalisatiON In chronic anorexia nervosa) [42]. However, a difference is the stepped-up, stepped-down philosophy as a main issue in FACT. FACT care then offers support in patients' own environment and, if needed, in patients' own home. Finally, FACT teams are composed of a variety of professionals, including psychiatrists, social workers and nurses, providing a broad range of expertise. It promotes a holistic approach where physical, psychological and social aspects remain central [44]. Given the complexity of mental health conditions and the discontinuity of care, coordinated care is indicated for persons with SEED-AN [44]. According to Delespaul (2013), persons with SEED-AN fulfil the criteria for SMI [45], and recent study findings underpin this statement [38]. Then, the (after)care for SEED-AN can be permanently improved by implementing active collaboration between the SEDU and FACT teams [18]. In practice, however, persons with SEED-AN are often excluded from FACT services because of the association between underweight and other serious health risks.

Despite the lack of exact numbers of presented SEED-AN patients benefiting from appropriate therapy in the Netherlands, it is assumed that not every SEED-AN patient receives appropriate support. This assumption is based on a small sample of the two mental health organisations involved in this research, the Parnassia Psychiatric Institute and Emergis, both located in the southwestern Netherlands. To explain this, a thorough look was taken at the following: (1) the current therapy offered by the SEDU programmes accessible to SEED-AN patients, and (2) the presence of the DSM-5 diagnosis of eating disorders in the patient bases of the FACT teams.

In the explanation of issue (1) Scrutinising the SEDU programmes revealed the following: There is an (outpatient) clinical therapy service for SEED-AN, provided within the network by Emergis. Ten beds are available, and outpatient (follow-up) care is provided for a period of up to 2 years. The treatment offered is supraregional; potentially, every Dutch SEED-AN patient can benefit from this treatment. However, this is a difficult issue; there is often a need for structural aftercare, which cannot always be provided because of distance. Then, issue (2) The sampling was carried out in the two mental health organisations involved. The DSM-5 code was searched within the patient base of the FACT teams. For privacy reasons, patient numbers and/or other personal characteristics that could be associated with individuals were not visible. The sample from Emergis included 2 people diagnosed with AN who received care at FACT Emergis. At the Parnassia Psychiatric Institute, samples were taken from two FACT care bases (PG-FACT Haaglanden and PG-FACT Rijnmond). The sample included 22 people diagnosed with comorbid AN (private administrative data, 2021). In total, 24 people diagnosed with AN were treated by FACT teams operating in the Dutch regions of Zuid-Holland and Zeeland in 2021. It is assumed that a substantial number of SEED-AN patients are not represented in FACT care.

This study aims to systematically examine the experiences of professionals and patients regarding existing care for SEED ANs across two settings in two health districts in the Netherlands. Namely - Specialised ED Unit and FACT.

The following research question was formulated: *Can,* and if so, how do FACT teams contribute to the treatment of patients suffering from SEED-AN from the perspective of professionals and SEED-AN patients?

Its objectives were as follows:

- Create a deeper understanding of how patients with SEED-AN already perceived support from the FACT service;
- To gain a deeper understanding of the beliefs and attitudes of both FACT professionals and SEDU professionals towards SEED-AN;
- Identify the perceived barriers experienced by both FACT and SEDU professionals in approaching patients suffering from SEED-AN;

# Method

This study was conducted between November 2021 and November 2023 in two participating mental health institutions in the Netherlands. Qualitative research makes it possible to give a voice to lived experiences; therefore, we conducted 22 semistructured interviews in focus groups and individually. Synthesis was then sought by combining and overlapping all sets of rich data constructed from both professional and patient respondents. Semistructured interviews were conducted and recorded by two lead researchers, a nurse practitioner and a nurse in training to become a nurse practitioner. The focus group interviews were conducted with professional respondents to discuss sensitive issues related to patients with SEED-AN in depth [46, p. 110]. Discussion was encouraged, and interaction among focus group participants broadened awareness of SEED-AN and related issues [46, p.110]. Given the vulnerability of the patient respondents, individual interviews were conducted. Rich and detailed data have been derived from the lived experiences of both patients and professionals [46, p. 80]. To closely reflect the realities, meanings, stories, interpretations and experiences of the respondents, reflexive thematic analysis (RTA) [47], which is supported by an inductive approach, was used to interpret the participants' views [46, p.174]. RTA is not used to discover the 'truth' [48], and we aimed to provide direction on the possibilities of an adapted approach for persons suffering from SEED-AN, where improving quality of life was paramount. Considering RTAs, both researchers were aware of the uncertainty that meaning was generated by the interpretation of the data and not by the amount of data excavated [49]. By mutual agreement of the researchers involved, it was decided to stop the data generation process as no longer fresh data appeared.

In RTAs, the researchers' positions, values and perspectives are integral. However, qualitative research is subjective. To ensure rigour, peer review and an independent research supervisor (for the nurse in training) and an independent researcher served as critical support [50]. We integrated several evaluation moments and used a logbook [51, p. 296]. To increase neutrality, both researchers were not involved in any of the treatments of the patients interviewed, and the researcher interviewing the professionals was not a direct colleague of the interviewees. Member checking increased reliability; respondents were presented with verbatim transcripts after the interview [52]. Transferability and authenticity were ensured by transcribing the interviews verbatim and describing the research context. Quotations from thick descriptions were used to reinforce the research findings [53, p. 505].

#### Sampling, recruitment and interviews

We applied purposive sampling [51, p. 190] to recruit professional respondents. The criteria:

- Professionals working in the FACT service;
- SEDU professionals.

For study participation, FACT staff and SEDU professionals were actively approached by our research team. Therefore, we used a strategy [51, p. 190]: the management of FACT and SEDU and the board of the participating institutes were invited in an online presentation to participate in the study. We highlighted the importance of improving care for SEED-AN patients and explained the study protocol. Managers were encouraged to participate in their staff for interviews. Our team contacted the FACT and SEDU practitioners personally by email or telephone to invite them to participate. We found sixteen professional respondents. Then, purposive sampling [51, p. 190] was applied in the recruitment of the patient respondents. The inclusion criteria were as follows:

- Diagnosis of anorexia nervosa, present for 5 years or intermittently present for 10 years or more;
- Patients who had received at least one usual treatment for AN (e.g., cognitive behavioural therapy enhanced (CBT-E [10]), clinical admission to a SEDU);
- Current treatment was received from a FACT professional in Parnassia.

Psychiatric Institute;

Not in mental health crisis (e.g., suicidal, psychotic).

To search for possible patient respondents, we used the business information of the Parnassia Psychiatric Institute and identified nineteen potential SEED-AN patient participants. The current therapists from the potential patient participants were then asked to test the established diagnostics to check if the diagnosis of AN was still accurate, and potential participants were then approached by their personal therapist for study participation. Thirteen candidates met the inclusion criteria. The exclusion criteria (being in crisis or incapacitated) were applied, resulting in six patients being excluded. Seven female patients agreed to participate and were interviewed. One interview was not used because the patient no longer met the criteria of SEED-AN [51, p. 184]. In conclusion, the patient sample consisted of six patient respondents.

# **Respondents' characteristics**

To protect confidentiality, a brief description of the characteristics of both professional respondents, representing professionals; two psychiatrists; four inpatient nurses; two clinical psychologists; three specialist psychiatric nurses; one nonverbal therapist (e.g., drama or physiotherapist); and four outpatient nurses, three male and thirteen female, is provided. It was expected that the variety of professionals was sufficient to answer the research question. All professional respondents were involved in the care of SEED-AN patients. Six professional respondents were employed in an SEDU; ten were FACT colleagues.

The characteristics of the patient respondents were as follows: all were female and aged between 26 and 65 years, with a mean age of 43 years. All patients were diagnosed with anorexia nervosa for at least 10 years and up to 52 years, with a mean duration of 27.5 years. All patients had received at least one eating disorder treatment in the past. All patients received tailor-made FACT treatment (e.g., pharmacotherapy, trauma therapy, emotional behavioral therapy, or a combination of both) for at least two years and up to thirteen years and received at least one evidence-based treatment for AN.

# Data collection and analysis

Data collection took place between September 2022 and April 2023. All professional respondents were interviewed by a senior researcher, and the patient respondents were interviewed by a nurse in training to become a nurse practitioner. Within the professional group, five focus groups were formed. Three focus groups consisted of FACT professionals only, and in two focus groups, a mix of SEDU and FACT professionals was presented. Then, a nurse practitioner, a community nurse, and a nonverbal therapist, all SEDU professionals, were interviewed individually due to agenda issues. The interviews started with guided questions from a topic list [53, p. 226]. Two topic lists were developed: one for interviewing professional respondents and one for interviewing patient respondents. Both lists were drafted and reviewed by two research colleagues (see Table 1, guided questions) [54, p. 185]. The topics concerning the professional respondents were drafted from the study proposal. The topics concerning the patient respondents were based on the six quality-of-life domains formulated by the World Health Organisation: physical health, mental health, level of independence, social relationships, environment, and personal values and beliefs [55].

The interviews were then transcribed verbatim and stored in a password-protected data software program. Then, in the analysis, the six-phase structure [47, 56] was used to define and refine the data: 1.) Familiarisation; (2.) coding; (3.) generating initial themes; (4.) developing and reviewing themes; (5.) refining themes; (6.) writing up, ultimately leading to a report. In addition, a semantic approach was adopted to stay close to the data.

 Table 1
 Guided questions of professional and patient respondents

Guided questions professional	Guided questions patient
respondents	respondents
What are your experiences in the treatment of SEED-AN?	Why you think you are receiving FACT care? For what purpose?
Do you think that FACT can contrib-	Do you receive any support in
ute to the treatment of SEED-AN	the area of giving meaning to
patients?	life and motivation? If so, how?
What do you think about the term	what has been your experience
SMI? Would the SEED-AN patient	in the treatment of your eating
meet the SMI criteria?	disorder?
Do you have an idea what kind of	Do you feel that the treatment
support is needed to treat SEED-AN	you receive is appropriate to
patients?	your problems and needs?
What kind of support would you need to treat SEED-AN patients?	What treatment options does FACT offer?
What do you think about a collabora- tion between the eating disorder department and FACT?	What are your expectations of your service providers?

# **Ethical considerations**

Prior to the study, permission was obtained from the Medical Ethics Committee of the Erasmus Medical Centre in Rotterdam. Permission was also sought and granted from the institutions involved (Emergis, Parnassia Psychiatric Institute) by the Scientific Committee of Emergis and Parnassia Psychiatric Institute.

# Results

Twenty-two respondents were informed about the purpose and procedure of the research, the voluntary nature of participation, the possibility of withdrawing from participation at any time during the interview, and all the respondents provided written consent for the use of their data. In total, we defined 4 themes, supported by 28 initial codes and 37 transcripts derived from rich data. The first 3 themes, A, "We feel ignorant in the treatment of SEED-AN patients", B, "There is a disability to act," and C, "We are more than executive practitioners alone," focus on rich data derived from professional respondents only. Theme D, The professionals at FACT have given me back my confidence in my treatment, is based on the data from patient respondents, see Fig. 1, illustrating the organisation of the interview transcripts, into initial codes, into the final themes.

# We feel ignorant in the treatment of SEED-AN patients

Across daily practice conditions, almost all the FACT professional respondents described feelings of "discouragement" in contact with other professionals when requests were made for the transfer of treatment. Some FACT professionals also experienced "distance from the SEED-AN person" and a "knowledge gap" as a result of preferring to address other mental health issues. Other FACT professionals were more active in the search for

support for mental health issues with which they were not familiar:

We have a knowledge gap. We need more eating disorder knowledge, which is also accessible to all FACT teams, and they have to make it easier to contact somebody at the SEDU, for instance, a name that is very important to us. (Professional respondents 1 & 2-FACT)

Next, all the professional SEDU respondents struggled with feelings of "discouragement" when trying to arrange follow-up care for their SEED-AN patients within FACT. One SEDU professional respondent explicitly referred to "loitering" in the attempt to organise cooperation between the SEDU and FACT, and one SEDU respondent stated "ignorance" and "no affinity" with the target group.

Loitering! FACT is then rejected, but there is no longer a direct treatment demand for individuals with eating disorders. An outpatient setting can provide only limited care. The WMO (Social Support Act in the Netherlands) is not adequate (professional respondent 14, SEDU).

Most of the FACT professional respondents experienced feelings of "resistance" against other psychiatric illnesses other than psychosis and, for example, serious mood disorders; they were more convinced that FACT was not providing appropriate treatment for SEED-AN patients, supported by their own personal preferences.

And you also choose for a long-term care patient, mostly those with psychotic disorders. With whom you just click. Yes, and I also think that this is the mechanism by which you get further away from other client groups. (Professional respondent 5-FACT)

Analysis then established a consensus among the FACT respondents on the concept of severe mental illness (SMI); according to all SEDU and FACT respondents, SEED-AN patients met the criteria for SMI.

When I look at my client why she was accepted for our program, she was rejected from other programs, and that criterion also meets the SMI (Prof. Resp. FACT 4).

If there is more comorbidity, then it is appropriate. Yes, then it would be SMI.

(Professional respondent 3 -FACT)

THEMES	INITIAL CODES	INTERVIEW TRANSCRIPTS
We feel ignorant in the treatment of SEED- AN patients	Loitering Distance Discouragement Resistance Knowledge gap Ignorance No affinity SMI	Loitering! FACT then rejects, while there is no longer a direct treatment demand in eating disorders. And an outpatient setting can only provide limited care. And WMO (Social Support Act in de Netherlands) is not adequate; We have a knowledge gap. We need more eating disorder knowledge, which is also accessible to all FACT teams, and they have to make it easier to contact somebody at the SEDU, for instance, a name; The FACT teams, which are also seen as some end-station; A psychotic person fits within our program, but other diagnoses had to be kept out a bit; And you also choose for a long-term care patient, mostly the psychotic disorders. With whom you just click. Yes, {} that this is how you get further away from other client groups; It's a bit blunt, but a lot of colleagues say to me: I just can't stand it, if someone doesn't eat, you just must eat; So, my conclusion: ignorance and no affinity with the target group; We want to be seen as a treatment department, not a gutter; I do see the same issues within other SMI individuals; she was rejected from other programs and that criterium also meets the SMI.
There is a disability to act	Failure Knowledge gap Solo work Vulnerability Discomfort Pressure	And in the beginning, {} I see progression, then, I have a bad feeling about the help offered by FACT only, and I prefer an expert on eating disorders and a GP involved. That requires cooperation and coordination, and it demands a lot from us as practical workers; I don't want to be trained anyway, because I hardly meet them (SEED-AN) in my practice; If there were more investment in psychoeducation and cooperation, and perhaps also meet the client together; I don't want to be the only one treating a SEED-AN individual, that's so vulnerable, unfortunately, I do not have a choice; You also must upskill our staff, but how do you motivate the staff to receive sufficient training? {} Mostly, the answer is, 'I'm not going to do the training anyway, because I hardly meet them in my practice; Training focuses mainly on increasing theoretical knowledge. But it is also about experiential knowledge. {} increasing skills. You cannot learn more about an eating disorder from theory alone; But none of us all have the expertise on eating disorders.
We are more than executive practitioners alone	Lack of vision Dissatisfaction Excessive workload Pressure Financial issues Thinking in boxes	As a practitioner, you need your management on board by explaining why treatment for SEED-AN is important; I also ask the management to focus: How do you want to organise your care in your institution? And I think that must be done differently; I sometimes feel a bit of pressure, to conclude treatment; Financial issues and how long someone has been in care should not be in the lead; The organisation of care for SEED-AN weighs so heavily on your already full agenda. Better financial support for indirect care (coordination of care) is also needed to get treatment funded; it is so time consuming to organise follow-up care; A mental health institution should have a leader with a certain 'out of the box' vision and leads towards that vision.
The professionals at FACT have given me back my confidence in my treatment.	Flexibility Accessibility Empathy Feeling seen Discussing patterns and ED behaviour Mirroring Reason to live Severe suffering	If my therapist is not available, I get a substitute; And there is the social service at home, or a crisis service I could call when needed; That you realize, I don't want to die, I want to live, so I restart eating; Improving life, yeah, that is a nice goal; A purpose in life is very important; Volunteering, getting among other people. Getting started; I'm seen as a person, not as a textbook. And that is a big thing; I am treated as a human being, not as a problem; It's also that my practitioner sees through me, that there's something going on with me, while I don't know it myself; If my therapist is not available, I get a substitute. And there is a housing service or crisis service that I could call when needed; When the eating disorder started, I couldn't believe it was just a symptom of other things going on. I learned that from this team; All those 9½ years I always had the same person, which was nice; It's every day, well, you get up and all that. But it's getting through the day, and going to bed at night, and thinking, another day is over.

#### **Disability to act**

Some FACT respondents described feelings of "discomfort" caused by the, sometimes, deplorable, physical condition of SEED-AN people, mostly related to 'working alone' within SEED-AN; one FACT respondent worried about the 'vulnerability' of their own position.

I do not want to be the only one treating a SEED-AN individual, that is so vulnerable; unfortunately, I do not have a choice. (Professional respondent 2-FACT)

One FACT respondent explicitly described the "pressure" of having to constantly coordinate and consult with multiple parties involved with the SEED-AN patient.

And in the beginning, sometimes, I see progression, then, I have a bad feeling about the help offered by FACT only, and I prefer an expert on eating disorders and a GP involved. That requires cooperation and coordination, and it demands a lot from us as practical workers. (Professional respondent 7, FACT)

However, some FACT and SEDU professional respondents explicitly reported increasing treatment skills and deepening their knowledge of SEED-AN's approach as a method to fight against a "knowledge gap" and feelings of "failure".

You also must upskill our staff, but how do you motivate the staff to receive sufficient training? {....} Mostly, the answer is, 'I'm not going to do the training anyway, because I hardly meet them in my practice; training focuses mainly on increasing theoretical knowledge. However, it is also about experiential knowledge. {....} increasing skills. You cannot learn more about EDs from theory alone. (Professional respondent 5- FACT)

# We are more than executive practitioners alone

All professional respondents expressed their "dissatisfaction" with the current mental health funding policy as a result of "financial decisions" made by the Dutch government and health insurers. Feelings of "pressure" and having an "excessive workload" were mentioned by almost all professional respondents.

The organisation of care for SEED-AN weighs so heavily on your already full agenda. Greater financial support for indirect care (coordination of care) is also needed to obtain treatment funding. In addition, it is time consuming to organise follow-up care. (Professional respondent 14, SEDU) Most professional respondents also experienced discrepancies in the chosen system of categorising mental health care. For example, there is too much thinking in "boxes". For example, for each mental health issue, a recovery pathway is developed, and professional respondents stated that these decisions are responsible for difficulties in the cooperation between care programs.

Financial issues and how long someone has been in care should not be in the lead. (Professional respondent 11, FACT)

Next, several of the SEDU and FACT professionals interviewed stated that there was a 'lack of vision' from management to provide treatment for SEED-AN patients.

As a practitioner, you need your management on board by explaining why treatment for SEED-AN is important. However, I also ask management to focus: How do you want to organise your care in your institution? And I think that must be done differently.

(Professional respondent 11-FACT).

I also sometimes feel a bit of pressure to conclude treatment (professional respondent-15, SEDU).

# The professionals at FACT have given me back my confidence in my treatment

All six patient respondents recognised "flexibility" and "accessibility" in the approach of their FACT therapists; three of them described several ways to contact their therapists, for example, by telephone, e-mail, or face-toface. If their personal therapist was not available, it was easy to contact a temporary substitute.

If my therapist is not available, I get a substitute. In addition, there is the housing service, or crisis service, I could call them when needed. (Patient respondent I)

All six patients described "empathy" in their therapists' approach and experienced trust in professional relationships, which is important in long-term professional relationships, as the patient respondents stated. Four patient respondents felt "seen" by their therapist, and one acknowledged the long-term professional relationship of 9.5 years, whereas one patient respondent explicitly preferred to have three therapists available, believing that the patient was too much of a burden for one person.

When I have a hard time in my head, I can turn to three people if I need to. In addition, that is nice,

because if it is one person I feel burdened by, then I feel I'm too much for the other person. (Patient respondent III)

With respect to eating behaviour issues, none of the patient respondents would interfere with the SEDU professionals; one patient respondent refused to "discuss eating behaviour" issues; and five recognised the ambivalence concerning discussing the eating disorder. Three patient respondents mentioned the need to confront and discuss eating disorder behaviour. "Mirroring", "discussing patterns", or "pointing out unconscious eating behaviour" were mentioned five times.

When the eating disorder started, I could not believe it was just a symptom of other things going on. I learned that from this team (Patient respondent III).

Finally, all patients' respondents stated issues such as a "reason to live" and improving their life by increasing social contact. All the patient respondents experienced new "goals in their life", initiated in cooperation with their FACT therapists. Some patient respondents explicitly did not want to die anymore; one patient respondent stated that life was "severe suffering".

It is every day, well, you get up and all that. However, it is getting through the day, and going to bed at night, and thinking, another day is over. (Pat. resp. V).

That you realise, I don't want to die, I want to live, so I restart eating (Patient respondent II).

# Discussion

The findings highlight how the care of SEED-AN patients is experienced in the participating mental health facilities. We used RTAs to develop themes from 22 interviews with SEDU and FACT professionals and persons suffering from SEED-AN.

Our findings resonate with previous qualitative research [18] and meet the SMI criteria [34, 38]. Interestingly, the FACT participants felt that the program met their expectations even though the providers lacked confidence. Survey research [57] (N=628) of patients with mental illness and substance use disorders and healthcare providers (N=471) highlighted stigmatisation during treatment. 52% of healthcare providers reported a focus on limitations and risk avoidance rather than opportunities and recovery, and 22% of patients reported a detached attitude from healthcare providers. The authors emphasise the importance of normalising mental health issues, which is an interesting point of view, as our findings highlight 'unfamiliarity with SEED-AN,' a 'knowledge gap', and 'lack of affinity in working with SEED-AN', which also resonates with previous research in the field of SEED-AN [58–60].

With these findings, we provide a deeper understanding of the beliefs and attitudes of both FACT professionals and SEDU professionals towards SEED-AN, and we identify some of the barriers experienced by the professionals involved in the SEED-AN approach.

Findings derived from health care providers highlight a wide range of policy-related issues. For example, the artificial distinction made by policymakers between the SEDU (curative) and the FACT teams (supportive), a high workload, and the request for improving financial policy and thus financial compensation for indirect patient contacts to enable consultation between the SEDU and FACT. However, the current healthcare landscape is challenging; in the Netherlands, 80,000 people are on the waiting list due to a large capacity shortage [61]. According to the Trimbos Institute (Dutch Research Institute for Mental Health), society is becoming increasingly complex; there is a structural outflow of mental health professionals of 10-15% per year, several mental health institutions are struggling to maintain their financial health, and strikingly, less attention is given to recovery, resulting in mentally ill patients remaining 'full-time' patients.

The findings highlight the consensus among professionals in the approach to the SEED-AN. Professionals consider SEED-AN a serious mental illness (SMI), and there is a need to address the multiple quality-of-life issues associated with SEED-AN. Our findings resonate with research comparing the Boston University psychiatric rehabilitation (BPR) approach with an active control condition (ACC) [38], the first study to include patients with a long-term eating disorder. Although not all experts agreed with the rehabilitation approach and stated a curative form, such as CBT-E, to minimise harm [62], professionals wish for better therapeutic methods where the preservation and/or improvement of the quality of life in persons living with SEED-AN should be the focus. This finding resonates with previous research conducted in the field of SEED-AN [18, 22, 58]. In our study, patients within FACT care recognised that, in their treatment, they were working towards, or had (partially) achieved, an improvement in quality of life. The patient respondents emphasised the importance of a close working relationship with the professional in which they wanted to be seen, known and recognised. This stated importance and desire for a therapeutic alliance within anorexia nervosa treatment has also been found in previous research [63, 64] and emphasises the importance of (restoring) epistemic trust within the therapeutic relationship, which refers to the ability to learn from new experiences on the basis of confidence that the knowledge gained is relevant,

reliable and applicable [64]. The desire and motivation to 'fight the eating disorder' seemed ambivalent in our study and variable in strength, which is in line with studies on SSCM [18, 32]. Perseverance was not constant, and the perceived level of suffering was high. To increase the therapeutic alliance among SEED-AN patients, Jochems [65] argued that by supporting the three basic psychological needs of autonomy, competence and relatedness, patients were better able to sustain the treatment process and engage in behavioural change, leading to better treatment outcomes [65]. Patient respondents expect professionals to confront, reflect and question them about food and weight, but taking the initiative to discuss these issues themselves is sometimes perceived as too difficult. It is therefore important that agreements about focus, goals, working methods and expectations are made before treatment begins [66]. All respondents value integrated care and collaboration with other healthcare providers, including those in the social sector. This requires good coordination and communication within health care networks [67]. FACT focuses on finding support within one's own network. With respect to patient respondents' outcomes, an adjustable model could benefit the cooperation of parties involved in the support of the patient. For example, the use of the resource group model. This model is ideal for realising collaboration between different parties, e.g., patients, FACTs, SEDUs, families and other important people [68], and shows similarities with the recovery-oriented approach [69]. The resource group model is a further development of assertive community treatment (ACT), created in the year 2000. A resource group is a group of people chosen by patients, consisting of professionals and nonprofessionals, who help achieve self-selected recovery goals [70]. Furthermore, using the resource group model, caregivers (e.g., SEDU and FACT professionals, family members) could easily exchange psychoeducation to receive a higher level of knowledge to support patient needs. We offer the model as a possible solution for the expressed needs of FACT and SEDU professionals for integrating each expertise and combining strengths to optimise SEED-AN treatment. The composition of the resource group, which is chosen by the patient and who can provide support, may have a positive impact on recovery and adherence to treatment, reducing the need for care and the risk of relapse [70]. An advantage of the resource group model is the steady implementation of the model in the Netherlands [71]. Further research in the field of SEED-AN that implements the resource group model is highly recommended.

# Conclusion

The findings highlight the impact of discouragement, the knowledge gap, the vulnerability of working solo within SEED-AN patients, and sometimes poor working conditions, as stated by the professional respondents, versus the experience of a close and long-term working relationship with their FACT therapists, as reported by SEED-AN patients. Our findings support the need for collaboration between patients, other key stakeholders, and various professionals to improve quality of life. The use of a resource model provides tools to improve this collaboration, and we highly recommend further research on the implementation of a resource group model in the care of patients suffering from SEED-AN.

# Limitations

In autumn 2020, a request was formulated to improve the treatment offered to patients suffering from SEED-AN. Although the study started vigorously, it slowed down during the COVID-19 pandemic. A staff shortage effectively reduced the availability of the research group. However, these obstacles did not affect the trustworthiness or validity of the results. The results were critically discussed and presented in an interim presentation (on paper and online). To obtain sufficient data, it was then decided to conduct individual interviews with professionals in addition to focus groups. With respect to the nurse in training, the training status requires that a nurse in training conducts research within his or her own mental health facility. Therefore, to achieve an average 'cross-sectional sample', a broad recruitment search was conducted within the Parnassia Psychiatric Institute only.

#### Abbreviations

SEED-AN Severe and enduring eating disorders - Anorexia nervosa AN Anorexia nervosa SEDU Specialised eating disorder unit FACT Flexible assertive community treatment PG Parnassia groep SMI Severe mental illness RTA Reflexive thematic analysis WMO Wet Maatschappelijke Ondersteuning (Social Support Act) BPR Boston University psychiatric rehabilitation ACC Active control condition

#### Author contributions

L.S. and M.G. wrote the complete manuscript. L.S. prepared all figures. All authors reviewed the manuscript.

#### Funding

Not applicable

#### Data availability

No datasets were generated or analysed during the current study.

#### Declarations

### **Consent for publication**

All authors have given their consent for publication of this manuscript.

#### Competing interests

The authors declare no competing interests.

#### Author details

<sup>1</sup>Parnassia Groep, YOUZ. Leggelostraat 85, 2541HR, The Hague, The Netherlands

<sup>2</sup>Parnassia Groep, PsyQ. Max Euwelaan 70, 3062 MA, Rotterdam, The Netherlands

<sup>3</sup>Parnassia Groep, PsyQ. Lijnbaan 4, 2512 VA, The Hague, The Netherlands
 <sup>4</sup>Emergis, Department of Eating disorders, Oostmolenweg 101,
 Kloetinge 4481 PM, The Netherlands

# Received: 31 May 2024 / Accepted: 4 December 2024 Published online: 24 December 2024

#### References

- Taquet M, Geddes JR, Luciano S, Harrison PJ. Incidence and outcomes of eating disorders during the COVID-19 pandemic. Br J Psychiatry. 2021;220(5):1–3. https://doi.org/10.1192/bjp.2021.105. Epub ahead of print. PMID: 35048812; PMCID: PMC7612698.
- van Eeden AE, van Hoeken D, Hoek HW. Incidence, prevalence and mortality of anorexia nervosa and bulimia nervosa. Curr Opin Psychiatry. 2021;34(6):515–24. https://doi.org/10.1097/YCO.00000000000739. PubMed PMID: 34419970; PubMed Central PMCID: PMC8500372.
- van Elburg AA, Danner UN. Anorexia nervosa and adolescents. Tijdschrift Voor Psychiatrie. 2016;57(12):923–7.
- Gutiérrez E, Carrera O, Severe, Enduring Anorexia Nervosa. Enduring Wrong Assumptions? Front Psychiatry. 2021;11:538997. https://doi.org/10.3389/fpsyt .2020.538997. PMID: 33658948; PMCID: PMC7917110.
- Smink FR, van Hoeken D, Hoek HW. Epidemiology of eating disorders: incidence, prevalence and mortality rates. Curr Psychiatry Rep. 2012;14(4):406– 14. https://doi.org/10.1007/s11920-012-0282-y. PubMed PMID: 22644309; PubMed Central PMCID: PMC3409365.
- Fichter MM, Quadflieg N, Crosby RD, Koch S. Long-term outcome of anorexia nervosa: results from a large clinical longitudinal study. Int J Eat Disord. 2017;50(9):1018–30. 22736. PubMed PMID: 28644530.
- Harris EC, Barraclough B. Excess mortality of mental disorder. Br J Psychiatry. 1998;173:11–53. PubMed PMID: 9850203.
- Brown C, Mehler PS. Medical complications of anorexia nervosa and their treatments: an update on some critical aspects. Eat Weight Disord. 2015;20(4):419–25. https://doi.org/10.1007/s40519-015-0202-3. PubMed PMID: 26138740.
- Steinhausen HC. The outcome of anorexia nervosa in the 20th century. Am J Psychiatry. 2002;159(8):1284–93. https://doi.org/10.1176/appi.ajp.159.8.1284. PubMed PMID: 12153817.
- Fairburn CG, Cooper Z, O'Connor M. The Eating Disorder Examination, 16th edition. In: Fairburn CG, editor. Cognitive behavior therapy and eating disorders. New York: Guilford Press; 2008. pp. 265–308.
- Guillaume S, Jaussent I, Olié E, Genty C, Bringer J, Courtet P, Schmidt U. Characteristics of suicide attempts in anorexia and bulimia nervosa: a case– control study. PLoS ONE. 2011;6(8):e23578. https://doi.org/10.1371/journal.po ne.0023578. PubMed PMID: 21858173; PubMed Central PMCID: PMC3155572.
- Marcolini F, Ravaglia A, Tempia Valenta S, Bosco G, Marconi G, De Ronchi D, Atti AR. Severe enduring anorexia nervosa (SE-AN) treatment options and their effectiveness: a review of literature. J Eat Disord. 2024;12(1):1006. https://doi.org/10.1186/s40337-024-01006-y. PMID: 38654374; PMCID: PMC11040982.
- Kiely L, Conti J, Hay P. Conceptualisation of severe and enduring anorexia nervosa: a qualitative meta-synthesis. BMC Psychiatry. 2003;23(1):606. https:// doi.org/10.1186/s12888-023-05098-9. PMID:37596588; PMCID: PMC10439651.
- Robinson PH, Kukucska R, Guidetti G, Leavey G. Severe and enduring anorexia nervosa (SEED-AN): a qualitative study of patients with 20 + years of anorexia nervosa. Eur Eat Disord Rev. 2015;23(4):318–26. https://doi.org/10.1002/erv.23 67. PubMed PMID: 26059633.
- 15. Akwa GGZ. Zorgstandaard eetstoornissen [Internet]. In herziening 2017. http s://www.ggzstandaarden.nl/zorgstandaarden/eetstoornissen/organisatie-va n-zorg/kwaliteitsbeleid/kwaliteitsstatuut [cited 2024 Nov 7].
- Tchanturia K, Lounes N, Holttum S. Cognitive remediation in anorexia nervosa and related conditions: a systematic review. Eur Eat Disord Rev. 2014;22(6):454–62. https://doi.org/10.1002/ecrv.2326. PubMed PMID: 25277720.
- 17. Tenwolde A. Lichamelijke Aspecten. In: Vandereycken W, Noordenbos G, red., editors. Handboek Eetstoornissen. 3 rd ed. Utrecht: De Tijdstroom; 2002.
- Schut L. One step forward, one step backward. An exploration of the perception of adult people as influenced by the diagnosis of severe and enduring anorexia nervosa. Thesis; 2019.

- Treasure J, Murphy T, Szmukler G, Todd G, Gavan K, Joyce J. The experience of caregiving for severe mental illness: a comparison between anorexia nervosa and psychosis. Soc Psychiatry Psychiatr Epidemiol. 2001;36(7):343–7. https://d oi.org/10.1007/s001270170039. PubMed PMID: 11606003.
- Strober M. Managing the chronic, treatment-resistant patient with anorexia nervosa. Int J Eat Disord. 2004;36(3):245 – 55. https://doi.org/10.1002/eat.2005
   PubMed PMID: 15478130.
- Striegel-Moore RH, DeBar L, Wilson GT, Dickerson J, Rosselli F, Perrin N, Lynch F, Kraemer HC. Health services use in eating disorders. Psychol Med. 2008;38(10):1465–74. https://doi.org/10.1017/S0033291707001833. PubMed PMID: 17976250; PubMed Central PMCID: PMC2965447.
- Touyz S, Le Grange D, Lacey H, Hay P, Smith R, Maguire S, Bamford B, Pike KM, Crosby RD. Treating severe and enduring anorexia nervosa: a randomized controlled trial. Psychol Med. 2013;43(12):2501–11. PubMed PMID: 23642330.
- Noordenbos G. Ontbrekende consensus over behandeldoelen en herstelcriteria voor eetstoornissen. De Psycholoog. 2010: 36–44.
- Lock J, Le Grange D. Family-based treatment: where are we and where should we be going to improve recovery in child and adolescent eating disorders. Int J Eat Disord. 2019;52(4):481–7. https://doi.org/10.1002/eat.22980. Epub 2018 Dec 6. PMID: 30520532.
- 25. Shomaker LB, Tanofsky-Kraff M, Matherne CE, Mehari RD, Olsen CH, Marwitz SE, Bakalar JL, Ranzenhofer LM, Kelly NR, Schvey NA, Burke NL, Cassidy O, Brady SM, Dietz LJ, Wilfley DE, Yanovski SZ, Yanovski JA. A randomized, comparative pilot trial of family-based interpersonal psychotherapy for reducing psychosocial symptoms, disordered-eating, and excess weight gain in at-risk preadolescents with loss-of-control-eating. Int J Eat Disord. 2017;50(9):1084–94. https://doi.org/10.1002/eat.22980. Epub 2017 Jul 17. PMID: 28714097; PMCID: PMC5759342.019;52.
- Koreshe E, Paxton S, Miskovic-Wheatley J, Bryant E, Le A, Maloney D, National Eating Disorder Research Consortium, Touyz S, Maguire S. Prevention and early intervention in eating disorders: findings from a rapid review. J Eat Disord. 2023;11(1):38. https://doi.org/10.1186/s40337-023-00758-3. PMID: 36899428; PMCID: PMC9999654.
- Gezondheidsraad. Preventie en vroege behandeling van eetstoornissen. Den Haag: Gezondheidsraad 2022; publicatienr. 2022/12 [internet]. https://www.g ezondheidsraad.nl/documenten/adviezen/2022/06/28/samenvatting-preven tie-en-vroege-behandeling-van-eetstoornissen. [cited 2024 May 24].
- Agras WS, Fitzsimmons-Craft EE, Wilfley DE. Evolution of cognitive-behavioral therapy for eating disorders. Behav Res Ther. 2017;88:26–36. https://doi.org/1 0.1016/j.brat.2016.09.004. PubMed PMID: 28110674; PubMed Central PMCID: PMC6136834.
- Lock J. An update on evidence-based psychosocial treatments for eating disorders in children and adolescents. J Clin Child Adolesc Psychol. 2015;44(5):707–21. PubMed PMID: 25580937.
- Treem J, Yager J, Gaudiani JL. A Life-Affirming Palliative Care Model for Severe and Enduring Anorexia Nervosa. AMA J Ethics. 2023;25(9):E703-709. https://d oi.org/10.1001/amajethics.2023.703. PMID: 37695873.
- McIntosh VV, Jordan J, Luty SE, Carter FA, McKenzie JM, Bulik CM, Joyce PR. Specialist supportive clinical management for anorexia nervosa. Int J Eat Disord. 2006;39(8):625–32. https://doi.org/10.1002/eat.20297. PubMed PMID: 16937382.
- Kiely L, Touyz S, Conti J, Hay P. Conceptualizing specialist supportive clinical management (SSCM): current evidence and future directions. J Eat Disord. 2022;10(1):32. https://doi.org/10.1186/s40337-022-00557-2. PubMed PMID: 35255984; PubMed Central PMCID: PMC8900319.
- Zhu J, Hay PJ, Yang Y, Le Grange D, Lacey JH, Lujic S, Smith C, Touyz S. Specific psychological therapies versus other therapies or no treatment for severe and enduring anorexia nervosa. Cochrane Database Syst Rev. 2023;8(8):CD011570. https://doi.org/10.1002/14651858.CD011570.pub2. PMID: 37610143; PMCID: PMC10445422.
- Schut L, van Mol M, Akkermans L. Langdurige anorexia nervosa: een verbetervoorstel in de behandeling. De Verpleegkundig Specialist. 2021;18(6):37–9.
- Wonderlich SA, Bulik CM, Schmidt U, Steiger H, Hoek HW. Severe and enduring anorexia nervosa: update and observations about the current clinical reality. Int J Eat Disord. 2020;53(8):1303–12. https://doi.org/10.1002/eat.23283
   Epub 2020 May 2. PMID: 32359125.
- Kiely L, Conti J, Hay P. Severe and enduring anorexia nervosa and the proposed terminal anorexia category: an expanded meta synthesis. Eat Disord 2024 Aug 4:1–32. https://doi.org/10.1080/10640266.2024.2379635. Epub ahead of print. PMID: 39099227.

- Mochizuki Y, Isobe T, Endo Y, Iio A, Takebayashi K, Yamasue H. A preliminary study of collaborative group intervention with recovered peer supporters for eating disorders: Analyses including comparisons between in-person and online sessions. Psychiatr Clin Neurosci. 2024; First published: 07 June 2024. https://doi.org/10.1002/pcn5.200
- Sanches SA, Swildens WE, Schaefer B, Moerbeek M, Feenstra TL, van Asselt ADI, Danner UN, van Weeghel J, van Busschbach JT. Effectiveness of the Boston University Approach To Psychiatric Rehabilitation in Improving Social Participation in people with severe Mental illnesses: a Randomized Controlled Trial. Front Psychiatry. 2020;11:571640. https://doi.org/10.3389/fps yt.2020.571640. PMID: 33173519; PMCID: PMC7538503.
- Delespaul JJ. (2013). Consensus over de definitie van mensen met een ernstige psychische aandoening (epa) en hun aantal in Nederland. Tijdschrift voor Psychiatrie. 2013; 55(6);427–438.
- van Veldhuizen JR. FACT: a Dutch version of ACT. Community Ment Health J. 2007;43(4):421–33. https://doi.org/10.1007/s10597-007-9089-4. Epub 2007 May 21. PMID: 17514502.
- Nielsen CM, Hjorthøj C, Nordentoft M, Christensen U. A qualitative study on the Implementation of Flexible Assertive Community Treatment - an Integrated Community-based treatment model for patients with severe Mental illness. Int J Integr Care. 2021;21(2):13. https://doi.org/10.5334/ijic.5540. PMID: 33981190; PMCID: PMC8086721.
- Russell J, Mulvey B, Bennett H, Donnelly B, Frig E. Harm minimization in severe and enduring anorexia nervosa. Int Rev Psychiatry. 2019;31(4):391–402. Epub 2019 May 10. PMID: 31074662.
- Birch E, Downs J, Ayton A. Harm reduction in severe and long-standing Anorexia Nervosa: part of the journey but not the destination-a narrative review with lived experience. J Eat Disord. 2024;12(1):140. https://doi.org/10.1 186/s40337-024-01063-3. PMID: 39267190; PMCID: PMC11396453.
- Cohen D. Een ernstige psychiatrische aandoening verkort de levensverwachting. Huisarts en wetenschap, 2015;58(1), 16–18. https://doi.org/10.1007 /s12445-015-0011-y.
- Delespaul JJ. (2013). Consensus over de definitie van mensen met een ernstige psychische aandoening (epa) en hun aantal in Nederland. Tijdschrift voor Psychiatrie. 2013; 55(6);427–438.
- Braun V, Clarke V. Successful qualitative research: a practical guide for beginners. Volume 80. London: SAGE; 2013. p. 174.
- Braun V, Clarke V. Using thematic analysis in psychology. Qualitative Res Psychol. 2006;3(2):77–101. https://doi.org/10.1191/1478088706qp063oa.
- Braun V, Clarke V. Reflecting on reflexive thematic analysis. Qual Res Sport Exerc Health. 2019;11(4):589–97. https://doi.org/10.1080/2159676X.2019.162 8806.0.
- Braun V, Clarke V. To saturate or not to saturate? Questioning data saturation as a useful concept for thematic analysis and sample-size rationales. Qual Res Sport Exerc Health. 2019;13(2):201–16. https://doi.org/10.1080/2159676X.201 9.1704846.
- Nowell LS, Norris JM, White DE, Moules NJ. Thematic analysis: striving to meet the trustworthiness criteria. Int J Qual Methods. 2017;16:1–13. https://doi.org /10.1177/1609406917733847.
- 51. Verhoeven N. Wat is onderzoek? Praktijkboek Voor methoden en technieken. Volume 296, 6th ed. Amsterdam: Boom; 2018. p. 298. 190.
- McKim C. Meaningful Member-Checking: a Structured Approach to Memberchecking. Am J Qualitative Res. 2023;7(2):41–52.
- 53. Polit DF, Beck CT. Nursing research: Generating and assessing evidence for nursing practice. Volume 505. LWW; 2020. p. 226.
- Mortelmans D. Handboek kwalitatieve onderzoeksmethoden. 3d ed. Leuven: Acco uitgeverijen; 2020. P. 185.

- 55. World Health Organization Quality of Life Group (WHOQOL). WHOQOL-2012. Geneva: WHO; 2012.
- Maguire M, Delahunt B. Doing a thematic analysis: a practical, step-by-step guide for learning and teaching scholars. AISHE-J. 2017;9(3):3351.
- 57. Van Erp N, Knispel A, Michon H, de Lange A, Hulsbosch L, Boumans J, Kroon H, van Weeghel J. Stigmatisering binnen de ggz: onderzoek onder cliënten en hulpverleners. Tijdschr Psychiatr. 2022;64(9):568–73.
- Schut L, Wright KM, Duckworth JE. Exploring quality of life in women with severe and enduring anorexia nervosa. Ment Health Pract. 2021. https://doi.org/10.7748/mhp.2021.e1587.
- 59. Easter MM. Not all my fault: Genetics, stigma, and personal responsibility for women with eating disorders. Soc Sci Med. 2012;75(8):1408–16.
- Puhl R, Suh Y. Stigma and eating and weight disorders. Curr Psychiatry Rep. 2015;17(3):552.
- 61. Trimbos-instituut. Hervorming ggz [Internet]. Utrecht: Trimbos-instituut; [cited 2024 Nov 29]. https://www.trimbos.nl/kennis/zorg-en-participatie/herv orming-ggz/
- 62. Dalle Grave R. Severe and enduring anorexia nervosa: no easy solutions. Int J Eat Disord. 2020;53(8):1320–1. https://doi.org/10.1002/eat.23295. Epub 2020 May 13. PMID: 32400903.
- De la Rie S, Noordenbos G, Donker M, Van Furth E. The quality of treatment of eating disorders: a comparison of the therapists' and the patients' perspective. Int J Eat Disord. 2008;41(4):307–17. https://doi.org/10.1002/eat.20494.
- Fonagy P, Allison E. The role of mentalizing and epistemic trust in the therapeutic relationship. Psychotherapy. 2014;51(3):372–80. https://doi.org/10.103 7/a0036505.
- Jochems E, Duivenvoorden HJ, Van Dam A, Mulder C. Motivation, treatment engagement, and psychosocial outcomes in outpatients with severe mental illness: a test of self-determination theory. Int J Methods Psychiatr Res. 2017;26(1). https://doi.org/10.1002/mpr.1537.
- Paulson-Karlsson G, Nevonen L. Anorexia nervosa: treatment expectations a qualitative study. J Multidiscip Healthc. 2012;5:169–77. https://doi.org/10.214 7/jmdh.s33658.
- Westen K, Van Vugt M, Bähler T, Mulder C, Delespaul P, Kroon H. FACT als onderdeel van wijkgerichte zorg in het (zorg)netwerk. Tijdschr Psychiatr. 2020;62(4):308–16.
- Leeman E, Tjaden C, Bovenberg F, Mulder C, Kienhorst G. Praktijkboek resourcegroepen: herstellen doe Je Samen. Amsterdam: Boom; 2021.
- Dawson L, Rhodes P, Touyz S. The recovery model and anorexia nervosa. Aust N Z J Psychiatry. 2014;48(11):1009–16. https://doi.org/10.1177/000486741453 9398. Epub 2014 Jun 13. PMID: 24927735.
- Tjaden C, Mulder C, Den Hollander W, Castelein S, Delespaul P, Keet R, et al. Effectiveness of resource groups for improving empowerment, quality of life, and functioning of people with severe mental illness: a randomized clinical trial. JAMA Psychiatry. 2021;78(12):1309–18. https://doi.org/10.1001/jamapsyc hiatry.2021.2880.
- Kenniscentrum Phrenos. Onderzoeksoverzicht december 2020 [Internet]. Utrecht: Kenniscentrum Phrenos. 2020 [cited 2024 Nov 29]. https://kenniscen trumphrenos.nl/assets/2021/07/Onderzoeksoverzichtdecember2020.pdf

# Publisher's note

Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.